

# Temi di Discussione

(Working Papers)

Performance pay and shifts in macroeconomic correlations

by Francesco Nucci and Marianna Riggi







# Temi di discussione

(Working papers)

Performance pay and shifts in macroeconomic correlations

by Francesco Nucci and Marianna Riggi

Number 800 - March 2011

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

*Editorial Board:* Marcello Pericoli, Silvia Magri, Luisa Carpinelli, Emanuela Ciapanna, Daniela Marconi, Andrea Neri, Marzia Romanelli, Concetta Rondinelli, Tiziano Ropele, Andrea Silvestrini. *Editorial Assistants:* Roberto Marano, Nicoletta Olivanti.

### PERFORMANCE PAY AND SHIFTS IN MACROECONOMIC CORRELATIONS

by Francesco Nucci<sup>\*</sup> and Marianna Riggi<sup>+</sup>

#### Abstract

A coincidence in time between the volatility break associated with the "Great Moderation" and large changes in the pattern of conditional and unconditional correlations between output, hours and labor productivity was detected by Galí and Gambetti (2009). We provide a novel explanation for these findings, based on the major changes that occurred in the U.S. design of labor compensation around the mid-1980s. These include a substantial increase in the incidence of performance pay coupled with a higher responsiveness of real wages to the business cycle. We capture this shift in the structure of labor compensation in a Dynamic New Keynesian (DNK) model and show that, by itself, it generates the disappearance of the procyclical response of labor productivity to non-technology shocks and a reduction of the contractionary effects of technology shocks on hours worked. Moreover, it accounts for a large share of the observed drop in output volatility after 1984 and for most of the observed changes in unconditional correlations.

JEL Classification: E24, E32, J3, J22.

Keywords: procyclical productivity, wage rigidities, performance pay.

1. Introduction	5
2. The Anatomy of the Great Moderation	
3. Theoretical Framework	
3.1 Households	11
3.2 Firms	13
3.3 Monetary Policy	16
3.4 The stationary representation of the model	
4 The role of Performance-related Pay	
4.1 Performance related pay and Changes in the Structure of Conditional Correlations	s 18
4.2 Performance pay and Changes in the Structure of Unconditional Second Moment	s 22
5. The role of hours adjustment costs	
6. Conclusions	
Appendix	
Tables and figures	
References	

## Contents

<sup>\*</sup> Sapienza University of Rome

<sup>&</sup>lt;sup>+</sup> Bank of Italy

# **1** Introduction<sup>1</sup>

Substantial changes in the correlation structure of U.S. macroeconomic series accompanied the downward shift in the volatility of output in the mid-1980s, which has been described as "the Great Moderation".<sup>2</sup> Galí and Gambetti (GG hereafter) (2009) estimate a structural vector autoregression (SVAR) with time-varying coefficients and provide evidence of a significant shift in the patterns of unconditional and conditional comovements among output, hours, and labor productivity as well as in the impulse responses to identified shocks.

Whether the volatility break and the large changes in the pattern of conditional and unconditional second moments experienced by the U.S. economy have a common underlying explanation is an open question. In this paper, we claim that some relevant developments in the design of labor compensation represent the common source of the drop in output volatility and the shift in the structure of conditional and unconditional correlations between output, hours and labor productivity.

The literature documents two major changes in the US wage setting at around mid-80s. The first one is an overall reduction in the degree of real wage rigidities. For instance, Blanchard and Galí (2007b) and Blanchard and Riggi (2009) emphasize the role of increased wage flexibility in accounting for the decrease in the macroeconomic effects of oil shocks. And Galí and van Rens (2010) detect a notable rise in the volatility of the real wage characterizing post-war U.S. dynamics, which contrasts with the parallel decline in output volatility.

<sup>&</sup>lt;sup>1</sup>We thank Olivier Blanchard, Efrem Castelnuovo, Luca Gambetti, Gian Maria Milesi-Ferretti, Salvatore Nisticò, Pau Rabanal, Giorgio Rodano, an anomimous referee and seminar participants at the Bank of Italy, the University of Rome Tor Vergata, the University of Padua and the CEPR-CREI Workshop on "Changes in Labor Market Dynamics" at Universitat Pompeu Fabra for helpful suggestions and discussions. Part of this project was completed while Marianna Riggi was a visiting scholar at the Research Department of the International Monetary Fund. The hospitality therein is gratefully acknowledged. The views expressed here do not necessarily reflect those of the Bank of Italy. Any errors are our own.

<sup>&</sup>lt;sup>2</sup>The Great Moderation was first documented by Kim and Nelson (1999) and McConnell and Perez-Quiros (2000), who estimate a break year of 1984 for the volatility of US GDP. An overview of the evidence and its explanations is provided by Blanchard and Simon (2001) and Stock of Watson (2002), who emphasize that the decline in volatility was experienced also in the other G7 countries.

The second striking fact is the substantial increase of firms' reliance on pay-for-performance mechanisms after the early 1980s. Lemieux et al. (2009), for example, using data from the Panel Study of Income Dynamics (PSID), document that the incidence of performance-pay jobs among salaried workers has risen from about 30 percent in the late 1970s to nearly 50 percent in the late 1990s. Moreover, on the basis of an annual survey of Fortune 1000 corporations, they calculate that the fraction of workers with some forms of performance-pay increased from 20.7 percent in 1987 to 44.5 percent in 2002. Mitchell et al. (1990) point to a significant shift towards adoption of incentive pay plans in the 1980s, after a period following the end of World War II to 1979 which was characterized by a decline in their use<sup>3</sup>. Importantly, the growing reliance on these compensation schemes has affected workers in general (Prendergast, 1999) and the increased share of performance-related pay in workers average earnings is associated with an increase in the estimated performance-pay sensitivity.<sup>4</sup>

We argue that the increased flexibility of real wages associated with a higher proportion of performance-related pay in workers average earnings may account for most of the results uncovered by GG (2009). We demonstrate this claim through a Dynamic New Keynesian (DNK) model which includes work effort as an additional dimension of firms' and households' choices. Using stochastic simulations, we find that, on its own, this shift in the design of workers pay yields the following implications:

- A vanishing procyclical response of labor productivity to non-technology shocks.
- A shrinking contractionary effects on hours worked of positive technol-

<sup>&</sup>lt;sup>3</sup>Similarly, Hall and Liebman (1998) show that the responsiveness of pay to performance for chief executive officers increased after the early 1980s. Jensen and Murphy (1990) estimate low pay-performance sensitivity for top executives in the 1970s to the mid 1980s.

<sup>&</sup>lt;sup>4</sup>Cuñat and Guadalupe (2005 and 2008) show that the increased responsiveness of pay to performance originates from the trend towards increased product market competition, that induced firms to re-shape the structure of the incentives to reward workers, making performance-related pay schemes more pervasive. Another explanation for the higher incidence of these pay systems is based on the extensive development of performance appraisal systems, due to progress in information and monitoring technologies which provided improved worker performance measures. Lemieux et al. (2009) discuss these advances and point to the extraordinary expansion over the last 30 years of consulting companies specializing in compensation.

ogy shocks.

- A dramatic decline in the procyclicality of labor productivity measured by a notable drop in the unconditional correlation between labor productivity and output as well as by a sign switch in the unconditional correlation with hours, from positive to large negative values.
- A sizeable drop in output volatility, i.e. a great part of the "Great Moderation".

These predictions closely correspond to the empirical findings of GG (2009) on the shift in the structure of conditional and unconditional second moments of U.S. macroeconomic series between the pre-1984 period and the post-1984 period.

By contrast, we show that if the higher wage responsiveness to the business cycle had been associated with a lower relative share of performance pay in workers average earnings, the changes in the patterns of conditional and unconditional correlations would have been exactly the opposite to those reported by GG (2009) in their empirical analysis.

We also investigate whether a structural change in the labor market through a generalized fall in the adjustment costs of labor, rather than through a change in the design of compensation, is conducive in our model to shifts in the pattern of comovements that resemble the empirical ones.<sup>5</sup> We show that while it may explain the lower procyclicality of labor productivity, it yields predictions that are at odds with the observed changes in the hours response to technology shocks. Of course, we do not rule out the potential relevance of this and other alternative explanations of the Great Moderation. However, our paper emphasizes that the developments occurred in the design of labor compensation, leading to a higher relevance of performance-related pay in the cyclical fluctuations of wages, have played a key role in explaining the enhanced macroeconomic stability and the parallel shift in the structure of macroeconomic correlations.

The paper is organized as follows. Section 1 provides evidence on the large changes in the structure of the correlations among output, hours and productivity associated with the Great Moderation. Section 2 presents the theoretical model. Section 3 discusses the implications of the change in the structure

<sup>&</sup>lt;sup>5</sup>This interpretation is suggested by Barnichon (2007), Galí and van Rens (2010) and GG (2009).

of labor compensation for the impulse responses to identified shocks, for the comovements among output, hours and labor productivity and for output volatility. Section 4 analyzes the macroeconomic consequences of a drop in labor adjustment costs and Section 5 concludes.

# 2 The Anatomy of the Great Moderation

GG (2008) analyze the correlation structure of U.S. output, labor productivity and hours over the Post-WWII period, and document a dramatic shift in the patterns of conditional and unconditional second moments of these macroeconomic series with the onset of the Great Moderation. Most of their evidence is obtained by estimating a SVAR with time-varying coefficients and stochastic volatility. The model is bivariate and includes (first differences of the logarithm of) labor productivity and (the log level of) hours per capita. Technology shocks are identified using long-run restrictions as in Galí (1999): the unit root in labor productivity originates uniquely in technology innovations so that only the latter have permanent effects on productivity. A number of empirical findings arise from their analysis and the major ones are the following.

**Changes in conditional correlations** The correlation of labor productivity with both output and hours *conditional* on technology and non - technology shocks shows large changes since the mid-1980s which are reflected in the impulse responses to identified shocks and can be summarized in two empirical results.

First, while in the pre-Great Moderation period labor productivity responds positively and significantly to non-technological innovations, displaying a pro-cyclical profile, in the more recent sub-period the response of labor productivity is negative and labor productivity becomes countercyclical conditional on non - technology shocks. Second, in the pre-Great Moderation period the effect of a technology shock on hours is negative and significant, while in the post-1984 period the effect, albeit still negative, is smaller (in absolute value) and almost reaching zero<sup>6</sup>. Figures 1a and 1b summarize

<sup>&</sup>lt;sup>6</sup>The effect on impact of technology shocks on labor input is a rather controversial issue in the literature. On empirical ground, Galí (1999) and Basu et al. (2006), among many others, document a negative correlation between labor productivity and hours conditional on technology shocks. They use different identification methods: long run restrictions

the bulk of this evidence. The first panel deals with the average responses of labor productivity to a non-technology shock in the pre-1984 and post-1984 periods. The second panel documents the average responses of hours to technology shocks in the two periods.

**Changes in unconditional correlations** As for the changes in unconditional correlations, GG (2009) use Band-Pass (BP) filtered data and report a dramatic fall in the cyclical behavior of labor productivity. The unconditional correlation between labor productivity and output becomes close to zero in the post-84 sample period while it was 0.61 in the pre-84 period. By using hours worked as the cyclical indicator, labor productivity becomes a countercyclical variable as the unconditional correlation between labor productivity and hours changes from 0.18 to -0.46. Consistently with GG, Barnichon (2007) finds that the sign of the correlation between cyclical unemployment and the cyclical component of labor productivity switched in the mid-80s: from negative it became positive. Stiroh (2009) confirms that the increased output stability in the US reflects a significant decline in the correlation between labor productivity growth and hours growth.

**Changes in volatility** As for the changes in unconditional volatilities, GG document that the standard deviation of output drops from 2.59 in the pre-84 period to 1.23 in the post-84 period and that the decline in the standard deviation of hours and labor productivity is not as large as that experienced by output. On the other hand, Galí and van Rens (2010) find

in the former case, a direct measure of technology change derived through a growth accounting approach in the latter case. The finding of a contractionary effect of technology improvements has been questioned along several dimensions. Christiano et al. (2003) argue that it originates from a specification error due to over-differencing of hours worked: if hours (per capita) are assumed to be a stationary variable and their level is used in the VAR analysis, then hours worked would rise, rather than fall, in the aftermath of a technology improvement. Chari et. al (2008) take the stand that Galí's (1999) result is potentially flawed by small-sample bias. In response to these criticisms, Fernald (2007) shows that once one allows for (statistically and economically plausible) trend breaks in productivity, the treatment of hours is relatively unimportant and hours fall sharply on impact following a technology improvement. Francis and Ramey (2005) provide arguments supporting the unit root hypothesis for per capita hours. Indeed, the impulse response functions from the model with stationary hours indicate that non-technology shocks have long-lived significant effects on labor productivity contrary to the fundamental identifying assumption.

that the standard deviation of both the product and consumption wages has increased by, respectively, 40 and 10 percent in the post-84 sample compared to the previous sub-period.

The complex picture that emerges from these findings challenges the "good luck" view of the Great Moderation and suggests that the US economy has witnessed more than a mere scaling down of the size of shocks. In what follows, we argue that the increased output stability and the shifts in the pattern of macroeconomic correlations have a common source which is related to the important changes occurred in the structure of labor pay.

# 3 Theoretical Framework

In this section we develop a DNK model with variable labor effort. Model features can be summarized as follows:

- Given our objectives, we assume that there are two sources of fluctuations in the economy: a technology and a non-technology shock. We characterize the nature of the shocks in accordance with the empirical analysis by GG, whose findings we seek to explain. In particular, we assume that the exogenous productivity factor has a unit root and is thus non-stationary in levels so that a technology shock affects productivity in the long-run. On the contrary, we assume a stationary process for what we call the non-technology shock and characterize it as a preference shock, as in Galí and Van Rens (2010).
- Effective labor input used by firms is a function of hours worked and effort exerted.
- Following Galí (1999), we split labor compensation in two components: the one rewarding hours and the one rewarding performance. Moreover, along the line of Blanchard and Galí (2007a and 2010), we introduce *ad hoc* rigidities in the labor compensation.
- We introduce hours adjustment costs, which, while not needed for the intuition behind our argument, allow us to shed light on how far we can go by assuming a more flexible labor market, which implies a decrease

in "labor hoarding", without any changes in the structure of labor compensation.<sup>7</sup>

## 3.1 Households

We consider a continuum of households uniformly distributed on the unit interval. Household j maximizes

$$E_{0} \sum_{t=0}^{\infty} \beta^{t} \xi_{t} \left\{ U[C_{t}(j), C_{t-1}] - g[H_{t}(j), E_{t}(j)] \right\},$$
(1)

where  $C_t(j)$  is household j's consumption in period t of the usual Dixit-Stiglitz aggregate of goods with elasticity of substitution  $\epsilon$ ,  $C_{t-1}$  is the aggregate consumption level in period t-1, the function g(.) measures the disutility from work, which depends on hours  $H_t$  and effort exerted  $E_t$  and  $\beta$  is the discount factor;  $\xi_t$  is a preference disturbance term with mean unity which follows the stationary first order autoregressive process

$$\log \xi_t = \rho_\xi \log \xi_{t-1} + \varepsilon_t^\xi, \tag{2}$$

where  $\varepsilon_t^{\xi}$  are zero-mean, i.i.d. innovations. We assume a specification of the period utility consistent with the balanced growth path:

$$U[C_t(j), C_{t-1}] \equiv \log[C_t(j) - hC_{t-1}], \qquad (3)$$

where  $h \in [0, 1)$  denotes the degree of external habit formation. Following Bils and Cho (1994) and Barnichon (2007), we assume that the period disutility of labor takes the form:

$$g(H_t(j), E_t(j)) = \frac{\lambda_h}{1 + \sigma_h} H_t^{1 + \sigma_h}(j) + H_t(j) \frac{\lambda_e}{1 + \sigma_e} E_t^{1 + \sigma_e}(j), \qquad (4)$$

where  $\lambda_h$ ,  $\lambda_e$ ,  $\sigma_h$ , and  $\sigma_e$  are positive constants.

The period budget constraint, conditional on the optimal allocation of expenditures among different goods, is given by:

$$P_t C_t + Q_t B_t = W_t H_t + V_t E_t + B_{t-1} + \Pi_t,$$
(5)

<sup>&</sup>lt;sup>7</sup>The presence of labor hoarding is a common explanation for the procyclicality of labor productivity and the ensuing short run increasing returns to labor (SRIRL) puzzle.

where  $P_t$  is the price level,  $Q_t$  is the price of a one-period nominally riskless bond, paying one unit of money and  $B_t$  is the amount of that bond purchased in period t. As in Galí (1999),  $W_t$  and  $V_t$  represent the nominal prices of one hour's work (nominal hourly wage hereafter) and of a unit of effort, respectively, where the latter represents our measure of performance pay.  $\Pi_t$ denotes the households' profits from ownership of firms.

The optimal consumption/savings and labor supply decisions are described by

$$Q_{t} = \beta E_{t} \frac{\xi_{t+1}}{\xi_{t}} \left( \frac{C_{t} - hC_{t-1}}{C_{t+1} - hC_{t}} \right) \frac{P_{t}}{P_{t+1}}$$
(6)

$$W_t^r \equiv \frac{W_t}{P_t} = (C_t - hC_{t-1}) \left( \lambda_h H_t^{\sigma_h} + \frac{\lambda_e}{1 + \sigma_e} E_t^{1 + \sigma_e} \right)$$
(7)

$$V_t^r \equiv \frac{V_t}{P_t} = (C_t - hC_{t-1}) \lambda_e H_t E_t^{\sigma_e}.$$
(8)

Equations (7) and (8) are the perfectly competitive hour and effort supply schedules, respectively. We relax the assumption of perfect competition in the labor market along the lines suggested by Blanchard and Galí (2007a, 2007b and 2010), by introducing ad hoc rigidities into the compensation structure, which prevent the cyclical components of both  $W_t^r$  and  $V_t^r$  from fully adjusting to their competitive counterparts. In particular, let us detrend  $W_t^r$  and  $V_t^r$  with the non-stationary level of technology,  $A_t$ , so that  $\widetilde{W}_t^r \equiv \frac{W_t^r}{A_t}$ and  $\widetilde{V}_t^r \equiv \frac{V_t^r}{A_t}$  denote the cyclical components of the hourly real wage and the real price of effort, respectively. We then replace equations (7) and (8) with the following hourly wage and performance pay curves, which, together, characterize the structure of labor compensation:

$$\widetilde{W}_{t}^{r} = \left[\frac{\left(C_{t} - hC_{t-1}\right)\left(\lambda_{h}H_{t}^{\sigma_{h}} + \frac{\lambda_{e}}{1+\sigma_{e}}E_{t}^{1+\sigma_{e}}\right)}{A_{t}}\right]^{1-\gamma_{h}},\qquad(9)$$

$$\widetilde{V}_t^r = \left[\frac{\left(C_t - hC_{t-1}\right)\lambda_e H_t E_t^{\sigma_e}}{A_t}\right]^{1-\gamma_e},\tag{10}$$

where the parameters  $\gamma_e$  and  $\gamma_h$  measure the sensitivity of worker's pay to the business cycle. The lower are  $\gamma_e$  and  $\gamma_h$  the higher is the responsiveness of labor compensation to economic conditions.

Changing  $\gamma_h$  and  $\gamma_e$ , however, changes both the degree of cyclicality of labor compensation as well as the stationarized steady state values of its two components:  $\overline{\widetilde{W^r}}$  and  $\overline{\widetilde{V^r}}$ . In general, an increase in the flexibility of labor compensation may leave unchanged the shares of performance pay and hourly pay in workers average earnings or, conversely, it may be associated with a change in the relative importance of the two compensation margins in workers average earnings. Indeed, for a given value of  $\gamma_h$ , a decrease in  $\gamma_e$  captures a higher flexibility of labor compensation associated with a higher relevance of performance pay in the cyclical fluctuations of labor compensation, i.e., with a higher share of effort-related pay in worker's average earnings,  $\left(\frac{\overline{\widetilde{V^r}}}{\overline{\widetilde{W^r}}+\overline{\widetilde{V^r}}}\right)$ . Conversely, for a given value of  $\gamma_e$ , a decrease in  $\gamma_h$  reflects a higher flexibility of labor compensation associated with a higher relevance of hourly pay in the cyclical fluctuations of labor compensation, i.e., with a higher share of hourly pay in worker's average earnings,  $\left(\frac{\widetilde{W^r}}{\widetilde{W^r}+\widetilde{V^r}}\right)$ . Thus, the parameters  $\gamma_e$  and  $\gamma_h$  affect the degree of responsiveness to economic conditions of the cyclical components of the two remuneration margins as well as their relative shares in workers average earnings.

## 3.2 Firms

We distinguish between two sectors: retail and wholesale firms. Households are employed by wholesale firms which face convex costs of varying their hours input and operate in a competitive market in relation to the goods they produce. Wholesale firms sell their output to retailers, which are monopolistically competitive and set prices in a staggered fashion, as in Calvo (1983). The separation of the two sectors is a modelling strategy which aims at providing a convenient separation of the two frictions (hours adjustment costs and price stickiness) in the model.

#### 3.2.1 Wholesale firms

Production by wholesale firm j is

$$Y_{it}^w = A_t L_{it}^\alpha, \tag{11}$$

where  $\alpha \in (0,1]$  and  $L_{jt}$  denotes the *effective* labor input defined as a function of hours and effort:<sup>8</sup>

$$L_{jt} = H_{jt} E_{jt},\tag{12}$$

and  $A_t$  was defined earlier as the productivity factor common across firms. We assume a trending nonstationary process for the technology level,  $A_t$ , to be consistent with the identification approach in GG which assumes that only technology shocks have a permanent effect on the level of productivity. Following Altig et al. (2005) and Gertler et al.(2008), we assume that  $Z_t \equiv \frac{A_t}{A_{t+1}}$  obeys the following exogenous stochastic process:

$$\log Z_t = (1 - \rho_a) \log \gamma_a + \rho_a \log Z_{t-1} + \varepsilon_t^a, \tag{13}$$

where  $\gamma_a$  defines the constant growth rate and  $\varepsilon_t^a$  is the i.i.d. technology shock.

Each firm j varies its hours input by facing convex costs, which are increasing with the speed of the desired adjustment and are measured in terms of the final good, according to the following adjustment cost function:

$$G_{jt} = \frac{\phi_h}{2} \left(\frac{H_{jt}}{H_{jt-1}} - 1\right)^2 Y_t,$$
(14)

where  $\phi_h \ge 0$  is the hours' adjustment cost parameter. The adjustment cost function implies that firms have an incentive to make gradual changes to hours which result in an intertemporal smoothing of their demand for hours.

Let  $\Pi_{jt}$  denote firm j' s period t profit. The wholesale firm maximizes

$$E_t \sum_{k=0}^{\infty} \beta^k \frac{\xi_{t+k}}{\xi_t} \left( \frac{C_t - hC_{t-1}}{C_{t+k} - hC_{t+k-1}} \right) \Pi_{jt+k}$$
(15)

subject to (11), where  $\Pi_{jt} = \frac{1}{\mu_t} Y_{jt}^w - \frac{W_t}{P_t} H_{jt} - \frac{V_t}{P_t} E_{jt} - \frac{\phi_h}{2} (\frac{H_{jt}}{H_{jt-1}} - 1)^2 Y_t$ and  $\mu_t = \frac{P_t}{P_t^w}$  is the markup of retail over wholesale prices. The first order conditions for this problem imply

$$V_t^r \equiv \frac{V_t}{P_t} = \frac{MPE_{jt}}{\mu_t} \tag{16}$$

<sup>&</sup>lt;sup>8</sup>See Galí (1999); as in the standard literature, we assume that effort is perfectly observable.

and

$$W_{t}^{r} \equiv \frac{W_{t}}{P_{t}} = \frac{MPH_{jt}}{\mu_{t}} - \phi_{h} \left(\frac{H_{jt}}{H_{it-1}} - 1\right) \frac{Y_{t}}{H_{jt-1}} + \beta E_{t} \frac{\xi_{t+1}}{\xi_{t}} \frac{C_{t} - hC_{t-1}}{C_{t+1} - hC_{t}} \phi_{h} \left(\frac{H_{jt+1}}{H_{jt}} - 1\right) \frac{H_{jt+1}}{H_{jt}^{2}} Y_{t+1}$$
(17)

where  $MPH_{j,t}$  is the marginal product of hours  $(MPH_{j,t} \equiv \alpha A_t H_{jt}^{\alpha-1} E_{jt}^{\alpha})$ and  $MPE_{j,t}$  is the marginal product of effort  $(MPE_{j,t} \equiv \alpha A_t H_{jt}^{\alpha} E_{jt}^{\alpha-1})$ . In symmetric equilibrium  $H_{j,t} = H_t$  and  $E_{j,t} = E_t$  for all j, since all wholesale firms are identical and make the same decisions.

#### 3.2.2 Retailers

We assume a continuum of monopolistically competitive retailers, indexed by i on the unit interval. The retail firm purchases the wholesale output and converts it into a differentiated final good, according to the following technology:

$$Y_t(i) = Y_t^w(i), \tag{18}$$

where  $Y_t^w(i)$  is the quantity of the (single) wholesale good.

Following Calvo (1983), retailers can reset their price at random dates: in each period only a randomly chosen fraction  $(1 - \theta)$  of retailers adjusts their prices. The remaining retailers keep their prices unchanged. Let  $P_t^*$ denote the new price set by a retailer at time t; then the aggregate price index satisfies

$$P_{t} = \left[\theta \left(P_{t-1}\right)^{1-\epsilon} + (1-\theta) \left(P_{t}^{*}\right)^{1-\epsilon}\right]^{\frac{1}{1-\epsilon}}.$$
(19)

The pricing decision of a retail firm obeys the following equilibrium condition:

$$E_{t} \sum_{k=0}^{\infty} \theta^{k} Q_{t,t+k} Y_{t+k/t} \left( \frac{P_{t}^{*}}{P_{t-1}} - \frac{\epsilon}{\epsilon - 1} M C_{t+k/t} \frac{P_{t+k}}{P_{t-1}} \right) = 0$$
(20)

where  $MC_{t+k/t}$  denotes the real marginal cost in t+k for a firm that last reset its prices in t and  $Q_{t,t+k}$  is the stochastic discount factor for nominal payoffs,  $Q_{t,t+k} \equiv \beta^k \frac{\xi_{t+k}}{\xi_t} \left( \frac{C_t - hC_{t-1}}{C_{t+k} - hC_{t+k-1}} \right) \frac{P_t}{P_{t+k}}$ . Assuming that hours' adjustment costs are distributed to the aggregate households, market clearing requires<sup>9</sup>

$$A_t L_t^{\alpha} = C_t \int_0^1 \left[ \frac{P_t(i)}{P_t} \right]^{-\epsilon} di.$$
(21)

## **3.3** Monetary Policy

The model is closed by a monetary policy rule. We assume that monetary policy obeys a simple Taylor rule, in which the nominal interest rate reacts to the current level of inflation and to the cyclical behavior of output:

$$\frac{1+i_t}{\overline{R}} = \pi_t^{\phi_\pi} \left(\frac{Y_t}{A_t}\right)^{\phi_y}$$

where  $\overline{R}$  is the steady state nominal gross rate and  $\phi_{\pi}$  and  $\phi_{y}$  are parameters.

## 3.4 The stationary representation of the model

The non-stationary technology process induces a stochastic trend in  $C_t$ ,  $Y_t$ ,  $W_t^r$  and  $V_t^r$ . To obtain the stationary representation of the model, the nonstationary variables are rescaled by the level of technology:  $\tilde{X}_t = \frac{X_t}{A_t}$ , where  $X_t$  is a generic variable and  $\tilde{X}_t$  its corresponding stationary ratio (see e.g. Gertler *et al.*, 2008, Juillard *et al.*, 2008 and Smets and Wouters, 2007). The detrended model is then log-linearized around the balanced growth steady state. The complete system of log-linear equations in stationary form is the following:

1. Technology

$$\widetilde{y}_t = \alpha \left( h_t + e_t \right)$$

2. Euler Equation

$$\widetilde{c}_{t} = \frac{\frac{\mu}{\gamma_{a}}}{1 + \frac{h}{\gamma_{a}}} (\widetilde{c}_{t-1} - z_{t}) + \frac{1}{1 + \frac{h}{\gamma_{a}}} E_{t} (\widetilde{c}_{t+1} + z_{t+1}) + \frac{1 - \frac{h}{\gamma_{a}}}{1 + \frac{h}{\gamma_{a}}} (i_{t} - E_{t} \pi_{t+1} + E_{t} \Delta \log \xi_{t+1})$$

<sup>9</sup>Market clearing for good *i* requires  $A_t L_{i,t}^{\alpha} = C_t(i)$ . Integrating over *i* yields  $A_t L_t^{\alpha} = \int_0^1 C_t(i) di = C_t \int_0^1 \frac{C_t(i)}{C_t} di = C_t \int_0^1 \left[\frac{P_t(i)}{P_t}\right]^{-\epsilon} di$ .

3. Performance Compensation

$$\widetilde{v}_t^r = (1 - \gamma_e) \left[ \frac{1}{1 - \frac{h}{\gamma_a}} \left( \widetilde{c}_t - \frac{h}{\gamma_a} \widetilde{c}_{t-1} + \frac{h}{\gamma_a} z_t \right) + h_t + \sigma_e e_t \right]$$

4. Hour Compensation

$$\widetilde{w}_{t}^{r} = (1 - \gamma_{h}) \left[ \frac{1}{1 - \frac{h}{\gamma_{a}}} \left( \widetilde{c}_{t} - \frac{h}{\gamma_{a}} \widetilde{c}_{t-1} + \frac{h}{\gamma_{a}} z_{t} \right) + \frac{\lambda_{h} \sigma_{h}}{\lambda_{h} + \frac{\lambda_{e}}{1 + \sigma_{e}}} \overline{E}^{1 + \sigma_{e}} h_{t} + \frac{\lambda_{e} \overline{E}^{1 + \sigma_{e}}}{\lambda_{h} + \frac{\lambda_{e}}{1 + \sigma_{e}}} \overline{E}^{1 + \sigma_{e}} e_{t} \right]$$

5. Phillips Curve

$$\pi_t = \beta E_t \pi_{t+1} + \frac{(1-\beta\theta)(1-\theta)}{\theta} \frac{\alpha}{\alpha + \epsilon(1-\alpha)} \left[ \widetilde{w}_t^r - \widetilde{y}_t + h_t + \frac{\epsilon}{\epsilon-1} \frac{\phi_h}{\alpha} \left( h_t - h_{t-1} \right) - \beta \frac{\epsilon}{\epsilon-1} \frac{\phi_h}{\alpha} \left( E_t h_{t+1} - h_t \right) \right]$$

- 6. Aggregate Resource Constraint
- $\widetilde{y}_t = \widetilde{c}_t$
- 7. Efficient Price Frontier
- $\widetilde{v}_t^r = \widetilde{w}_t^r e_t + h_t + \frac{\epsilon}{\epsilon 1} \frac{\phi_h}{\alpha} \left( h_t h_{t-1} \right) \beta \frac{\epsilon}{\epsilon 1} \frac{\phi_h}{\alpha} \left( E_t h_{t+1} h_t \right)$
- 8. Monetary policy rule

$$i_t = \phi_\pi \pi_t + \phi_y \widetilde{y}_t$$

In the following section, we conduct stochastic simulations in order to investigate the implications for macroeconomic correlations of a structural change in the design of labor compensation.

# 4 The role of Performance-related Pay

This section establishes that the higher flexibility of labor compensation, coupled with a higher share of performance-related pay in workers average earnings, does explain most of the empirical evidence summarized in section 2. In order to pin down the role of changes in the US design of labor compensation, we fix all the model's parameters except  $\gamma_e$  and investigate if an increasing relevance of performance-related pay in the cyclical fluctuations of labor compensation can account for the observed shifts associated with the Great Moderation. It is important to emphasize that, in doing so, we do not maintain that the US economy has not undergone relevant structural changes other than those occurred in the structure of labor pay. By keeping

all the parameters but  $\gamma_e$  fixed across periods, we try to ascertain the specific ability of changes in labor pay in accounting for the shifting patterns of macroeconomic correlations and volatilities.

In line with what is widely accepted in the literature, we calibrate the discount factor,  $\beta$ , to a value of 0.99 and the elasticity of substitution among differentiated goods,  $\epsilon$ , to a value of 6, which is consistent with a gross steady state markup of 20 percent. The sticky price parameter,  $\theta$ , is calibrated equal to 0.6, which implies an average price duration of three quarters, a value consistent with the empirical evidence (see Sbordone, 2006). The degree of external habit formation h is set equal to 0.6. As for the monetary policy rule, we set  $\phi_{\pi} = 1.2$ , in the range of values (1, 5] that cover the empirically plausible set conditional on having a unique equilibrium, and  $\phi_{y} = 0.1$ . We calibrate the elasticity of output with respect to effective labor input,  $\alpha$ , to a value of 2/3, and the labor adjustment cost parameter,  $\phi_h$ , is set equal to 1. We assume that the elasticity of the marginal disutility of effort with respect to effort is equal to 1 ( $\sigma_e = 1$ ) and we calibrate  $\sigma_h$  to 1, in accordance with Schor's (1987) estimate of 0.5 for the elasticity of effort with respect to hours,  $\frac{\sigma_h}{1+\sigma_e}$ . The steady state balanced growth term  $\gamma_a$  is calibrated equal to 1.004, consistent with the average quarterly output growth of 0.4 per cent recorded in the U.S. data. Following Altig et al. (2005), Gertler et al. (2008) and Juillard et al. (2008) we assume a trending nonstationary AR(2) process for the level of technology,  $A_t$ , by setting  $\rho_a = 0.5$ . The nontechnology disturbance term is assumed to follow a stationary AR(1) process characterized by a first order autocorrelation parameter,  $\rho_{\xi}$ , equal to 0.9.

## 4.1 Performance related pay and Changes in the Structure of Conditional Correlations

#### The response of labor productivity to non-technology shocks

Figure 2a highlights that the response at impact (i.e. in the first period) of labor productivity to a non-technology shock is monotonically strictly increasing in  $\gamma_e$  and this holds true for any given value of  $\gamma_h$ . Hence, an increase in the responsiveness of the real wage to economic conditions associated with a higher relevance of the performance-related component in the cyclical fluctuations of labor compensation lowers the procyclicality of labor productivity conditional on non-technology shocks. When the share of performance-related pay is sufficiently large, labor productivity becomes

countercyclical.

Conversely, figure 2b shows that high values of  $\gamma_h$  are associated with countercyclical responses of labor productivity conditional on non-technology shocks and labor productivity becomes procyclical as  $\gamma_h$  approaches to zero. The response at impact of labor productivity to non-technology shocks is therefore monotonically strictly decreasing with  $\gamma_h$ , regardless of the values assigned to  $\gamma_e$ . In other words, an increase in the flexibility of labor compensation associated with a lower relative share of performance pay would imply an increase in the degree of procyclicality of labor productivity conditional on non-technology shocks.

The economic intuition behind this result is as follows. The apparent feature of short run increasing returns to labor (SRIRL) derives from the fact that firms vary the intensity of their labor utilization over the cycle. The extent to which this happens depends on the short run profitability of substituting increases in effort for increases in hours of work. Profitability depends on the structure of labor compensation and in particular on the shares of the two compensation margins: following an increase in demand, the smaller the increase in the compensation for effort, the more profitable it will be to react to current and expected changes in the economy by varying labor effort and the higher will be the degree of procyclicality of labor productivity.

To see this, consider the efficient price frontier:

$$\widetilde{v}_t^r - \widetilde{w}_t^r = h_t - e_t + \frac{\epsilon}{\epsilon - 1} \frac{\phi_h}{\alpha} \left( h_t - h_{t-1} \right) - \beta \frac{\epsilon}{\epsilon - 1} \frac{\phi_h}{\alpha} \left( E_t h_{t+1} - h_t \right) \quad (22)$$

Combining the efficient price frontier with the aggregate production function yields the following expression for the cyclical component of labor productivity:

$$\widetilde{mph}_{t} = \alpha \left( \widetilde{w}_{t}^{r} - \widetilde{v}_{t}^{r} \right) + (2\alpha - 1) \left( h_{t} \right) + \frac{\epsilon}{\epsilon - 1} \phi_{h} \left( h_{t} - h_{t-1} \right) - \beta \frac{\epsilon}{\epsilon - 1} \phi_{h} \left( E_{t} h_{t+1} - h_{t} \right)$$
(23)

In the aftermath of a positive non-technology shock, both effort and hours compensation tend to increase and, of course, so do hours. The previous equation therefore highlights that the sign of the response of labor productivity,  $\widetilde{mph}_t$ , to a non-technology shock depends on the relative movements of the two compensation margins,  $\widetilde{w}_t^r$  and  $\widetilde{v}_t^r$ . Labor productivity can indeed show a positive response to an expansionary non-technology shock if hours wages rise by more than the performance compensation. In other words, the sign and the magnitude of labor productivity response depends crucially on the interplay between the variations of the hourly wage and the performance pay. Thus, the increase in the relevance of performance pay in the cyclical fluctuations of labor compensation (i.e. a drop in  $\gamma_e$ ) can indeed account for the disappearance of the positive correlation between labor productivity and hours conditional on non-technology shocks, that occurred in the early 1980s.

#### The response of hours to technology shocks

Figure 3a shows that (the absolute value of) the response at impact of hours worked to technology improvements is monotonically strictly increasing in  $\gamma_e$ , irrespective of the value assigned to the parameter  $\gamma_h$ . Conversely, Figure 3b highlights that (the absolute value of) the response at impact of hours worked to technology improvements is monotonically strictly decreasing in  $\gamma_h$ , regardless of the value taken by the parameter  $\gamma_e$ .

The economic intuition is straightforward. The model predicts a negative response of firms' use of labor input. Indeed, the negative response of labor input to a technology expansion is theoretically consistent with a sticky price economy in which monetary policy is not fully accommodative (Gali 1999, Basu et al. 2006, Dotsey, 2002 and Galí and Rabanal, 2004).<sup>10</sup> Following a technology shock, both effort and hours compensation tend to rise. The efficient price frontier (22) implies that hours contraction will be larger the higher is the increase in hourly wages (i.e. the lower is  $\gamma_h$ ; see Riggi, 2010 and Riggi and Tancioni, 2010) and the smaller is the increase in performance compensation (i.e. the higher is  $\gamma_e$ ). Thus, for a given value of  $\gamma_h$ , an increase in the flexibility of labor compensation associated with an increase in the relevance of performance pay (i.e. a lower  $\gamma_e$ ) drives down the (absolute value of the) negative correlation between hours and productivity conditional to a technology shock, because it induces firms to reduce labor input by adjusting on the performance side more than on the hours side.

<sup>&</sup>lt;sup>10</sup>Following a technology improvement aggregate demand does not grow as much as it would under price flexibility and therefore the more productive firms are able to satisfy demand with fewer hours worked by employees.

#### A calibration exercise

A natural question emerges about the required variation in  $\gamma_e$  necessary to account for the changes in the structure of conditional correlations documented by GG (2009). The literature does not provide much guidance about setting a value for  $\gamma_e$  and even less so for calibrating a value to the pre-1984 period and another one for the post-1984 period. We therefore perform a calibration exercise to find the values of  $\gamma_e$  that minimize the distance between the theoretical, model-based IRFs and the empirical SVAR-based average IRFs of labor productivity and hours to non-technology and technology shocks, respectively. The empirical IRFs are those estimated by GG (2009) and depicted in Figures 1a and 1b. We set  $\gamma_h = 0.7$  in order to capture the high degree of labor compensation rigidities documented for the pre-1984 period (Blanchard and Riggi, 2009), and we maintain the calibration discussed above for the remaining parameters. The values of  $\gamma_e$  that provide the best approximation of the empirical SVAR-based average IRFs are  $\gamma_e$  = 0.719 in the pre-1984 sample and  $\gamma_e = 0.274$  in the post-1984 sample<sup>11</sup>. Given the calibration for the other parameters this implies an increased relevance of performance pay in workers average earnings of more than 16 percent.<sup>12</sup> Figures 4a and 4b report the theoretical IRFs.

This exercise highlights that the developments occurred in the design of labor compensation, leading to a higher relevance of performance related pay in the cyclical fluctuations of wages, go a long way in accounting for the changes in the pattern of conditional comovements among output, hours, and labor productivity. Consistent with the empirical findings, our model predicts that an increase in the performance pay sensitivity implies a change from positive to negative of the sign of labor productivity response to nontechnology innovations and a drop in absolute value of the negative response of hours to technology shocks.

<sup>&</sup>lt;sup>11</sup>The standard deviation of the estimated value of  $\gamma_e$  in the pre-1984 is 0.0212. The standard deviation of the estimated value of  $\gamma_e$  in the post-1984 is 0.0196.

<sup>&</sup>lt;sup>12</sup>Indeed, given the calibration of the other parameters, the two values of  $\gamma_e$  imply an increase in  $\frac{\widetilde{V^r}}{\widetilde{W^r} + \widetilde{V^r}}$  from to 0.432 to 0.503. See the Appendix on steady state calculations.

## 4.2 Performance pay and Changes in the Structure of Unconditional Second Moments

Stiroh (2009) and GG argue that a substantial fraction of the decline in output volatility characterizing the Great Moderation period can be explained by the relevant decline in the correlation between labor productivity and hours. In order to analyze the implications of the changes in the structure of labor compensation for macroeconomic unconditional correlations and volatilities, we use artificial data generated from our model economy in which both technology and non-technology shocks are present. In particular, we consider the two values of  $\gamma_e$  obtained in the previous calibration exercise: 0.719 and 0.274. For each parameterization, we extract randomly 1,000 samples of 100 observations each from the artificial dataset, compute second moments and average them across the 1,000 samples.<sup>13</sup> To appraise the ability of the increased performance pay sensitivity to account for the observed changes in macroeconomic correlations and volatilities, we keep the remaining parameters fixed at the calibrated values previously described in this section.

The first panel of table 1 shows the evidence by GG(2009) on the observed shift in macroeconomic correlations between output, labor productivity and hours. The second panel of table 1 reports the variations in the cyclical comovements between output and hours, hours and productivity, and output and productivity computed on the artificial data generated within our model by assuming that the change in  $\gamma_e$  had been the sole structural change occurred in the economy. The comparison between the model-based changes in cross correlations and the corresponding observed changes reported in the first panel reveals that the structural change on which we are focusing can account for the shifts in the comovements among macroeconomic variables. The correlation between output and hours is stable across the two subsamples, as documented by GG. Conversely, the procyclical behaviour of labor productivity seems to vanish when we turn to the post-1984 sample. Indeed, when we take output as a cyclical indicator, in the second sub-sample labor productivity becomes an essentially acyclical variable. In particular, GG document that the unconditional correlation between labor productivity and business output drops to 0.03 from a value of 0.61. In our analysis we document that, when  $\gamma_e$  takes a lower value, such correlation drops to 0.03

<sup>&</sup>lt;sup>13</sup>The sample length of 100 quarters is empirically reasonable since it broadly resembles the period associated with the Great Moderation (from 1984:1 onwards) and the pre-Great Moderation period (from 1948:1 to 1983:4).

(from a value of 0.95). By taking hours worked as a cyclical indicator labor productivity becomes a contercyclical variable. Indeed, GG report that the actual correlation shifts from 0.18 to -0.46 and our model predicts a shift from 0.90 to -0.38.

In table 2 we compare the evidence provided by GG and Galí and van Rens (2010) on the changes in the volatility of output and other variables with our results from artificial data when we vary the parameter  $\gamma_e$ . On its own, the increased performance sensitivity of pay accounts for a sizeable part of the drop in output volatility documented in the literature on the Great Moderation. GG report that the standard deviation of B-P filtered output declines by about 53 per cent in the post-1984 period and our model predicts a reduction of 22 percent when  $\gamma_e$  is set at the lower value.<sup>14</sup> GG also report a decline in the volatility of labor productivity and hours but they emphasize that it is less pronounced than that of output. Our model-based data, on the contrary, indicate that the standard deviation of labor productivity and hours has increased, rather than decreased (although to a small extent in the case of hours).<sup>15</sup>

The increase in the volatility of labor compensation in real terms, driven by the higher performance sensitivity of pay and measured by a shift in the standard deviation from 1.82 to 2.63, is consistent with the empirical evidence reported in Galí and Van Rens (2010) for the US economy, which shows a sizeable increase in the volatility of actual (real) wages over the post-1984 period. When they consider compensation per hour deflated using the compensation deflator from the BLS Labor Productivity and Cost (LPC) program and transformed using the B-P filter on the (log of) variable, the

<sup>&</sup>lt;sup>14</sup>In our model economy, we calibrate the standard deviation of the two shocks so as to obtain - in the high  $\gamma_e$  case - a standard deviation of output that is exactly equal to the actual standard deviation of output in the pre-1984 sample.

<sup>&</sup>lt;sup>15</sup>The increase in the unconditional volatility of hours worked is explained by the contribution of non-technology shocks. If the compensation for effort becomes less rigid (i.e.  $\gamma_e$  shrinks), while the compensation for hours is as rigid as before (i.e.  $\gamma_e$  stays unchanged), then following a demand shock it is profitable for firms to react by adjusting on the hours margin to a larger extent than before. This of course implies that the volatility of hours is expected to rise. By contrast, the increase in the unconditional volatility of labor productivity is mainly driven by the contribution of technology shocks. If the compensation for effort becomes less rigid, in the aftermath of a technology shock the variation (in absolute value) of performance pay relatively to the variation of hourly wage tends to be higher than before and this induces a smaller adjustment of hours worked and a larger variation of worker's performance. This implies an increase in the volatility of labor productivity.

standard deviation shifts from 0.71 to 0.99 for the pre- and post-1984 periods. We believe that this finding ought to be emphasized as it stands in contrast with the overall, parallel decline of macroeconomic volatility.<sup>16</sup>

## 5 The role of hours adjustment costs

A common explanation for the procyclical behavior of labor productivity is labor hoarding, which is caused by a variety of costs involved in adjusting the labor force. Since the latter cannot be costlessly adjusted in the short run, firms react to changes in demand by varying the intensity of labor utilization. Thus, the labor force is smoothed over the cycle, and the cyclical variations in labor effort, which increases in booms and decreases in recessions, generate a perception of SRIRL (Sbordone, 1996). It follows that a straightforward interpretation of the decline in the correlation between labor productivity and hours, conditional on non-technology shocks, would be based on the decrease of labor adjustment costs over the recent decades, due to higher competition, a more flexible labor market and tougher corporate governance. By developing a theoretical model, Galí and van Rens (2010) point to the decline in labor market frictions as the major source of the vanishing procyclicality of labor productivity.

In our model economy, this interpretation is captured by a decrease in the value of the parameter  $\phi_h$ . The role of  $\phi_h$  in shaping the procyclicality of labor productivity is highlighted by equation (23) which establishes that higher labor adjustment costs renders the SRIRL outcome more likely. Accordingly, Figure 5a shows that a downward shift of  $\phi_h$ , taking up different values ranging from 2 to zero, yields a gradual disappearance of the procyclical response of labor productivity to non-technology shocks. Indeed, the response at impact of labor productivity to a non-technology shock is monotonically strictly increasing in  $\phi_h$ , regardless of the parameterization used for  $\gamma_e$ .

However, the drop in labor hoarding driven by lower adjustment costs

 $<sup>^{16}</sup>$ As a robustness inspection of the results documented in Tables 1 and 2, we have verified whether the values of the unconditional moments computed on the artificial data may depend on initial conditions. To tackle this issue we have eliminated the first 100 observations from the overall sample of 10000 observations and extracted randomly, as before, 1,000 samples of 100 observations each from this revised artificial dataset. The results remain virtually unchanged.

would lead to an even larger initial drop in hours following a technology innovation, in contrast with the evidence in GG and the results of our SVARbased IRFs. Figure 5b shows that the response at impact of hours worked to technology improvements is, in general, monotonically strictly declining (in absolute value) in the adjustment cost parameter,  $\phi_h$ . Intuitively, hours response to a technology shock would be lower the higher are the hours adjustment costs.

Of course the inability of the decrease in labor adjustment costs to account on its own for both changes in the dynamic responses to shocks does not rule out the potential relevance that decreasing labor market frictions may have had for the Great Moderation. For example, the drop in labor adjustment costs combined with a stronger anti-inflationary stance of the monetary policy could make a good job in explaning the Great Moderation and the parallel shifts in macroeconomic correlations.

# 6 Conclusions

Galí and Gambetti (2009) show that large changes in the patterns of conditional and unconditional correlations among output, hours and labor productivity have accompanied the substantial decline in macroeconomic volatility experienced by the US economy since the mid-1980s.

In this paper we have proposed a novel explanation for these observed patterns based on extensive evidence showing that the U.S. design of labor compensation has changed around the mid-1980s, towards an increased flexibility of workers' pay combined with a greater relevance of the performancepay component in the cyclical fluctuations of labor compensation.

Using stochastic simulations, we have showed that a structural change towards an increased flexibility in real wages associated with a higher share of the performance-related component in workers compensation can account for the following empirical patterns: a) the disappearance of the procyclical productivity puzzle; b) the smaller contractionary effects on hours of a technology improvement; c) the decline in the unconditional correlation between labor productivity and both output and hours; d) the decline in output volatility and e) the parallel increase in the volatility of real wage.

We have also showed that if the higher wage flexibility had been accompanied by a *lower* share of the performance-related component in workers compensation we would have implications for the patterns of comovements which go in the opposite direction to those that emerge from empirical analysis.

The evidence by GG and Galí and van Rens (2010) is useful to shed light on the merits of different explanations for the Great Moderation. We do not claim that the shift in the design of labor compensation is the sole structural change behind the complex picture associated with the Great Moderation. However, our goal is to emphasize that the developments occurred in the design of labor compensation, leading to a higher relevance of performancerelated pay in the cyclical fluctuations of wages, have played a key role in explaining the enhanced macroeconomic stability and the parallel shift in the structure of macroeconomic correlations.

On the other hand, the explanation of shifts in macroeconomic correlations based on modifications to pay settings is particularly appealing as it can be reconciled with an empirical puzzle highlighted by Davis and Kahn (2008), namely that the substantial slump in the volatility of aggregate real activity has been coincident with an increase in *individual* income volatility and earnings uncertainty (see also Comin and Mulani, 2006). Arguably, these opposite trends in micro and macro volatility are potentially consistent with a higher incidence of the performance-related component of labor compensation, that rewards idiosyncratic effort and individual worker characteristics.

We therefore see two natural directions for future research. First, an extension to the model by allowing for heterogeneous consumers to investigate the coincidence between changes in the structure of macroeconomic correlations and the increased volatility of earnings at the individual level; and second, a structural estimation of the model.

#### Appendix

#### Steady state calculation

As it is standard in the literature, we normalize  $\overline{H}$  to be equal to one (see, e.g., Smets and Wouters, 2007). In order to compute the performance pay share on workers labor compensation, we need to recover the steady state value of  $\overline{E}$ . To this aim, we combine the cost minimization condition with (9) and (10) in the main text:

$$\overline{E} = \frac{\overline{\widetilde{W^r}}}{\overline{\widetilde{V^r}}} = \frac{\left[\left(\overline{\widetilde{C}} - \frac{h}{\gamma_a}\overline{\widetilde{C}}\right)\left(\lambda_h + \frac{\lambda_e}{1+\sigma_e}\overline{E}^{1+\sigma_e}\right)\right]^{1-\gamma_h}}{\left[\left(\overline{\widetilde{C}} - \frac{h}{\gamma_a}\overline{\widetilde{C}}\right)\lambda_e\overline{E}^{\sigma_e}\right]^{1-\gamma_e}}$$
(1A)

Using  $\overline{\widetilde{C}} = \overline{E}^{\alpha}$ , (1A) can be rewritten as follows:

$$\frac{\left[\lambda_{h} + \frac{\lambda_{e}}{1 + \sigma_{e}}\overline{E}^{1 + \sigma_{e}}\right]^{1 - \gamma_{h}}\overline{E}^{\alpha(1 - \gamma_{h})}}{\left(\lambda_{e}\overline{E}^{\sigma_{e}}\right)^{1 - \gamma_{e}}\overline{E}^{\alpha(1 - \gamma_{e})}}\left(1 - \frac{h}{\gamma_{a}}\right)^{\gamma_{e} - \gamma_{h}} = \overline{E}$$
(2A)

Rearranging yields:

$$\frac{\left(\lambda_h + \frac{\lambda_e}{1 + \sigma_e}\overline{E}^{1 + \sigma_e}\right)^{1 - \gamma_h}}{\lambda_e^{1 - \gamma_e}} \left(1 - \frac{h}{\gamma_a}\right)^{\gamma_e - \gamma_h} = \overline{E}^{1 - \alpha(\gamma_e - \gamma_h) + \sigma_e(1 - \gamma_e)}$$
(3A)

By solving the polynomial equation, we recover the steady state value of  $\overline{E}$ , which depends on  $\gamma_e$  (among the other parameters) and we are able to study how the performance pay share in workers labor compensation varies with  $\gamma_e$ . Indeed, the relative share of performance pay is equal to

$$\frac{\overline{\widetilde{V^r}}}{\overline{\widetilde{W^r}} + \overline{\widetilde{V^r}}} = \frac{1}{\overline{E} + 1}$$
(4A)

or, equivalently, looking at the share from the supply side:

$$\frac{\overline{\widetilde{Vr}}}{\widetilde{Wr}+\widetilde{Vr}} = \frac{\left(\widetilde{C}-\frac{h}{\gamma}\widetilde{C}\right)^{1-\gamma_e}\lambda_e^{1-\gamma_e}E^{\sigma_e(1-\gamma_e)}}{\left(\widetilde{C}-\frac{h}{\gamma}\widetilde{C}\right)^{1-\gamma_e}\lambda_e^{1-\gamma_e}E^{\sigma_e(1-\gamma_e)} + \left(\widetilde{C}-\frac{h}{\gamma}\widetilde{C}\right)^{1-\gamma_h}\left(\lambda_h+\frac{\lambda_e}{1+\sigma_e}E^{1+\sigma_e}\right)^{1-\gamma_h}} = \frac{1}{1+\frac{E^{\alpha}(\gamma_e-\gamma_h)-\sigma_e(1-\gamma_e)\left(1-\frac{h}{\gamma}\right)^{\gamma_e-\gamma_h}\left(\lambda_h+\frac{\lambda_e}{1+\sigma_e}E^{1+\sigma_e}\right)^{1-\gamma_h}}{\lambda_e^{1-\gamma_e}}} \tag{5A}$$

The value of the share obtained by plugging the model-consistent value of  $\overline{E}$  into(4A) coincides with the value obtained by plugging  $\overline{E}$  into (5A). We are thus able to recover the change in the ratio of performance pay to workers labor compensation which is consistent with the variation in  $\gamma_e$ , suggested by the minimum distance exercise presented in section 3.1 (main text).

Figure 1a The average responses of labor productivity to non-technology shocks



Figure 1b The average responses of hours to technology shocks



Source: our replication of Galì and Gambetti (2009)'results (see figures 6.B and 7.B on pages 47-48 of their paper).









Figure 4a. Nontechnology Shocks: Labor Productivity response



Figure 4b. Technology Shocks: Hours response



Evidence from GG (2009)				Second moments of artificial					
	(table 3, pp. 31)			model-based data					
	(BP-filter)			(cyclical components)					
	Pre-1984	Post-1984	Change	$\gamma_e = 0.719$	$\gamma_e = 0.274$	Change			
Output, hours	0.89	0.86	-0.02	0.99	0.91	-0.08			
Hours, productivity	0.18	-0.46	-0.65	0.90	-0.38	-1.28			
Output, productivity	0.61	0.03	-0.58	0.95	0.03	-0.92			

Table 1 Changes in Cross Correlations

# Table 2 Changes in Volatility

Evidence from GG and Galí-Van Rens (2010) S tan dard Deviation (BP-filter)			Second moments of artificial model-based data (cyclical components)			
	Pre-1984	Post-1984	$\frac{\text{Post-1984}}{\text{Pre-1984}}$	$\gamma_e = 0.719$	$\gamma_e=0.274$	$\frac{\text{Post-1984}}{\text{Pre-1984}}$
Output	2.59	1.23	0.47	2.59	2.02	0.78
Hours	2.08	1.39	0.67	1.94	2.19	1.12
Productivity	1.18	0.68	0.57	0.69	0.87	1.26
Wage	0.71	0.99	1.38	1.82	2.63	1.44



Figure 5. The role of hours adjustment costsFIGURE 5a. Non technology shocksFIGURE 5b. Technology shocksResponse at impact of labor productivityResponse at impact of hours

## References

- Altig, David, Lawrence Christiano, Martin Eichenbaum, Jasper Linde, (2005) "Firm-Specific Capital, Nominal Rigidities and the Business Cycle," NBER Working Paper 11149.
- [2] Barnichon, Regis (2007), "Productivity, Aggregate Demand and Unemployment Fluctuations," CEP Discussion Paper No 819.
- [3] Basu, Susanto, John G. Fernald and Miles S. Kimball (2006), "Are Technology Improvements contractionary?", American Economic Review, vol. 96(5), 1418-1448.
- [4] Bils Mark and Jang-Ok Cho (1994), "Cyclical factor utilization," Journal of Monetary Economics vol. 33(2), 319-354.
- [5] Blanchard J. Olivier and Jordì Galí (2007a), "Real Wage Rigidities and the New Keynesian Model", *Journal of Money, Credit and Banking*, supplement to vol. 39, n.1, 35-66.
- [6] Blanchard J. Olivier and Jordì Galí (2007b), "The Macroeconomic Effects of Oil Shocks: Why are the 2000s So Different from the 1970s?" NBER Chapters, in International Dimensions of Monetary Policy, pages 373-421 National Bureau of Economic Research, Inc.
- [7] Blanchard J. Olivier and Jordi Galí (2010), "Labor Markets and Monetary Policy: a New Keynesian Model with Unemployment," *American Economic Journal: Macroeconomics*, vol. 2(2), pages 1-30, April.
- [8] Blanchard J. Olivier and Marianna Riggi, (2009), "Why are the 2000s so different from the 1970s? A structural interpretation of changes in the macroeconomic effects of oil prices", NBER Working Paper 15467.
- [9] Blanchard, Olivier J. and John Simon, (2001), "The Long and Large Decline in Output Volatility," Brookings Papers on Economic Activity, issue 1, 135-164.
- [10] Calvo, Guillermo (1983), "Staggered Prices in a Utility maximizing Framework", Journal of Monetary Economics 12, 383-398.

- [11] Chari, Varadarajan V., Patrick J. Kehoe and Ellen R. McGrattan, (2008), "Are structural VARs with long-run restrictions useful in developing business cycle theory?," *Journal of Monetary Economics*, vol. 55(8), pages 1337-1352.
- [12] Christiano, Lawrence J.; Martin, Eichenbaum and Robert, J., Vigfusson, (2003) "What happens after a technology shock?," NBER Working Paper no. 6400.
- [13] Comin, Diego and Sunil Mulani (2006), "Diverging Trends in Aggregate and Firm Volatility", *Review of Economics and Statistics*, 88(2), 374-383.
- [14] Cuñat, Vicente and Maria Guadalupe (2005), "How does Product Market Competition Shape Incentive Contracts?", Journal of the European Economic Association, 3(5), 1058-1082
- [15] Cuñat, Vicente and Maria Guadalupe (2008), "Globalization and the Provision of Incentives Inside the Firm", *Journal of Labor Economics*, forthcoming.
- [16] Davis, Steven J. and James A. Kahn (2008), "Interpreting the Great Moderation: Changes in the Volatility of Economic Activity at the Macro and Micro Levels", *Journal of Economic Perspectives*, 22, 155-80.
- [17] Dotsey, Michael, (2002), "Structure from shocks," Federal Reserve Bank of Richmond Economic Quarterly, 88, 37-47.
- [18] Fernald, John G., (2007). "Trend breaks, long-run restrictions, and contractionary technology improvements," *Journal of Monetary Economics*, vol. 54(8), pp. 2467-2485.
- [19] Francis, Neville and Valery A. Ramey (2005). "Is the Technology-Driven Real Business Cycle Hypothesis Dead? Shocks and aggregate fluctuations revisited", *Journal of Monetary Economics*, 52 (8), 1379-1399.
- [20] Galí Jordi, (1999), "Technology, Employment, and the Business Cycle: Do Technology Shocks Explain Aggregate Fluctuations?", American Economic Review, 89, 249-271.

- [21] Galí Jordi and Luca Gambetti, (2009), "On the Sources of the Great Moderation" forthcoming in American Economic Journal: Macroeconomics, 1, 26-57.
- [22] Galí, Jordi and Pau Rabanal, (2004), "Technology Shocks and Aggregate Fluctuations: How Well Does the RBC Model Fit Postwar U.S. Data?" in NBER Macroecomics Annual, 2004, ed. Mark Gertler and Kenneth Rogoff, 225-318, Cambridge, MA: MIT Press.
- [23] Galí Jordi and Thijs Van Rens, (2010), "The Vanishing Procyclicality of Labor Productivity and the Great Moderation", mimeo.
- [24] Gertler, Mark., Sala, Luca and Antonella Trigari, (2008) "An Estimated Monetary DSGE Model with Unemployment and Staggered Nominal Wage Bargaining" *Journal of Money, Credit and Banking*, vol. 40 (8): 1713-1764.
- [25] Hall, Brian J., and Jeffrey B. Liebman (1998), "Are CEOs Really Paid Like Bureaucrats?," *Quarterly Journal of Economics*, CXIII, 653-691.
- [26] Jensen Michael C. and Kevin J. Murphy (1990), "Performance Pay and Top-Management Incentives", *The Journal of Political Economy*, Vol. 98, No. 2 (Apr., 1990), pp. 225-264
- [27] Juillard, Michael, Michael Kumhof, Ondra Kamenik, Douglas Laxton (2008) "Optimal price setting and in‡ation inertia in a rational expectations model" *Journal of Economic Dynamics and Control* 32(8), 2584-2621.
- [28] Kim, Chang-Jin and Charles R. Nelson (1999), "Has the U.S. Economy Become More Stable? A Bayesian Approach Based on a Markov Switching Model of the Business Cycle," *The Review of Economics and Statistics*, 81 (4) 608-616.
- [29] Lemieux, Thomas, MacLeod, W. Bentley and Parent, Daniel, (2009), "Performance Pay and Wage Inequality", *Quarterly Journal of Economics*, 124, 1, pp. 1-49.
- [30] McConnell, Margaret M. and Gabriel Perez-Quiros (2000), "Output Fluctuations in the United States: What Has Changed Since the Early 1980.s?," American Economic Review, vol. 90, no. 5, 1464-1476

- [31] Mitchell Daniel, Lewin David and Edward Lawler III (1990), "Alternative Pay system, Firm Performance and Productivity" in Blinder, Alan, (ed.) "Paying for Productivity: A Look at the Evidence", Washington, DC: The Brookings Institution.
- [32] Prendergast Canice, (1999), "The Provision of Incentives in Firms", Journal of Economic Literature, Vol. 37, No. 1. (Mar.,), pp. 7-63.
- [33] Riggi, Marianna (2010), "Nominal and real wage rigidities in New Keynesian models: a critical survey", *Journal of Economic Surveys*, vol. 24 Issue 3, November 2010, Pages 539-572.
- [34] Riggi, Marianna and Massimiliano Tancioni (2010), "Nominal v. Real Wage Rigidities in New Keynesian Models with Hiring Costs: a Bayesian Evaluation", *Journal of Economic Dynamics and Control*, vol. 34, Issue 7, pages 1305-1324.
- [35] Sbordone, Argia M. (1996), "Cyclical productivity in a model of labor hoarding", Journal of Monetary Economics, 38, 331-361.
- [36] Sbordone, Argia M. (2006), "U.S. Wage and Price Dynamics: A Limited-Information Approach", International Journal of Central Banking, vol. 2, No. 3, 155-191
- [37] Schor, Juliet, (1987), "Does Work Intensity Respond to Macroeconomic Variables?: Evidence from British Manufacturing, 1970-86", Mimeo.
- [38] Smets, Frank and Rafael, Wouters, (2007), "Shocks and Frictions in US Business Cycles: A Bayesian DSGE Approach", American Economic Review, 97, 3, 586-606.
- [39] Stiroh, Kevin J. (2009), "Volatility Accounting: A Production Perspective on Increased Economic Stability", Journal of the European Economic Association vol. 7 issue 4(6) 671-696.
- [40] Stock, James, and Mark W.Watson (2002), "Has the Business Cycle Changed and Why?", NBER Macroeconomics Annual 2002, MIT Press.

#### RECENTLY PUBLISHED "TEMI" (\*)

- N. 776 Measuring the price elasticity of import demand in the destination markets of *Italian exports*, by Alberto Felettigh and Stefano Federico (October 2010).
- N. 777 Income reporting behaviour in sample surveys, by Andrea Neri and Roberta Zizza (October 2010).
- N. 778 *The rise of risk-based pricing of mortgage interest rates in Italy*, by Silvia Magri and Raffaella Pico (October 2010).
- N. 779 On the interaction between market and credit risk: a factor-augmented vector autoregressive (FAVAR) approach, by Roberta Fiori and Simonetta Iannotti (October 2010).
- N. 780 Under/over-valuation of the stock market and cyclically adjusted earnings, by Marco Taboga (December 2010).
- N. 781 Changing institutions in the European market: the impact on mark-ups and rents allocation, by Antonio Bassanetti, Roberto Torrini and Francesco Zollino (December 2010).
- N. 782 Central bank's macroeconomic projections and learning, by Giuseppe Ferrero and Alessandro Secchi (December 2010).
- N. 783 (Non)persistent effects of fertility on female labour supply, by Concetta Rondinelli and Roberta Zizza (December 2010).
- N. 784 Stars and comets: an exploration of the patent universe, by Carlo Menon (January 2011).
- N. 785 Sectoral money demand and the great disinflation in the US, by Alessandro Calza and Andrea Zaghini (January 2011).
- N. 786 Public sector efficiency and political culture, by Raffaela Giordano and Pietro Tommasino (January 2011).
- N. 787 Monetary incentives vs. monitoring in addressing absenteeism: experimental evidence, by Francesco D'Amuri (January 2011).
- N. 788 *FaMIDAS: A Mixed Frequency Factor Model with MIDAS structure*, by Cecilia Frale and Libero Monteforte (January 2011).
- N. 789 Policies for local development: an evaluation of Italy's "Patti Territoriali", by Antonio Accetturo and Guido de Blasio (January 2011).
- N. 790 Testing for east-west contagion in the European banking sector during the financial crisis, by Emidio Cocozza and Paolo Piselli (February 2011).
- N. 791 Are incentives for R&D effective? Evidence from a regression discontinuity approach, by Raffaello Bronzini and Eleonora Iachini (February 2011).
- N. 792 Evaluating the impact of innovation incentives: evidence from an unexpected shortage of funds, by Guido de Blasio, Davide Fantino and Guido Pellegrini (February 2011).
- N. 793 Credit availability and investment in italy: lessons from the "Great Recession", by Eugenio Gaiotti (February 2011).
- N. 794 Bridging the gap between migrants and the banking system, by Giorgio Albareto and Paolo Emilio Mistrulli (February 2011).
- N. 795 I fondi comuni aperti in italiaItalia: performance delle società società di gestione del risparmio, by Michele Leonardo Bianchi and Maria Grazia Miele (February 2011).
- N. 796 *Securitization is not that evil after all*, by Ugo Albertazzi, Ginette Eramo, Leonardo Gambacorta and Carmelo Salleo (February 2011).

<sup>(\*)</sup> Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- P. ANGELINI, *Liquidity and announcement effects in the euro area*, Giornale degli Economisti e Annali di Economia, v. 67, 1, pp. 1-20, **TD No. 451 (October 2002).**
- P. ANGELINI, P. DEL GIOVANE, S. SIVIERO and D. TERLIZZESE, *Monetary policy in a monetary union: What role for regional information?*, International Journal of Central Banking, v. 4, 3, pp. 1-28, **TD No.** 457 (December 2002).
- F. SCHIVARDI and R. TORRINI, *Identifying the effects of firing restrictions through size-contingent Differences in regulation*, Labour Economics, v. 15, 3, pp. 482-511, **TD No. 504 (June 2004).**
- L. GUISO and M. PAIELLA,, *Risk aversion, wealth and background risk*, Journal of the European Economic Association, v. 6, 6, pp. 1109-1150, **TD No. 483 (September 2003).**
- C. BIANCOTTI, G. D'ALESSIO and A. NERI, *Measurement errors in the Bank of Italy's survey of household income and wealth*, Review of Income and Wealth, v. 54, 3, pp. 466-493, **TD No. 520 (October 2004).**
- S. MOMIGLIANO, J. HENRY and P. HERNÁNDEZ DE COS, The impact of government budget on prices: Evidence from macroeconometric models, Journal of Policy Modelling, v. 30, 1, pp. 123-143 TD No. 523 (October 2004).
- L. GAMBACORTA, *How do banks set interest rates?*, European Economic Review, v. 52, 5, pp. 792-819, **TD No. 542 (February 2005).**
- P. ANGELINI and A. GENERALE, On the evolution of firm size distributions, American Economic Review, v. 98, 1, pp. 426-438, **TD No. 549 (June 2005).**
- R. FELICI and M. PAGNINI, *Distance, bank heterogeneity and entry in local banking markets*, The Journal of Industrial Economics, v. 56, 3, pp. 500-534, No. 557 (June 2005).
- S. DI ADDARIO and E. PATACCHINI, *Wages and the city. Evidence from Italy*, Labour Economics, v.15, 5, pp. 1040-1061, **TD No. 570 (January 2006).**
- S. SCALIA, *Is foreign exchange intervention effective?*, Journal of International Money and Finance, v. 27, 4, pp. 529-546, **TD No. 579 (February 2006).**
- M. PERICOLI and M. TABOGA, Canonical term-structure models with observable factors and the dynamics of bond risk premia, Journal of Money, Credit and Banking, v. 40, 7, pp. 1471-88, TD No. 580 (February 2006).
- E. VIVIANO, *Entry regulations and labour market outcomes. Evidence from the Italian retail trade sector*, Labour Economics, v. 15, 6, pp. 1200-1222, **TD No. 594 (May 2006).**
- S. FEDERICO and G. A. MINERVA, Outward FDI and local employment growth in Italy, Review of World Economics, Journal of Money, Credit and Banking, v. 144, 2, pp. 295-324, TD No. 613 (February 2007).
- F. BUSETTI and A. HARVEY, *Testing for trend*, Econometric Theory, v. 24, 1, pp. 72-87, **TD No. 614** (February 2007).
- V. CESTARI, P. DEL GIOVANE and C. ROSSI-ARNAUD, Memory for prices and the Euro cash changeover: an analysis for cinema prices in Italy, In P. Del Giovane e R. Sabbatini (eds.), The Euro Inflation and Consumers' Perceptions. Lessons from Italy, Berlin-Heidelberg, Springer, TD No. 619 (February 2007).
- B. H. HALL, F. LOTTI and J. MAIRESSE, Employment, innovation and productivity: evidence from Italian manufacturing microdata, Industrial and Corporate Change, v. 17, 4, pp. 813-839, TD No. 622 (April 2007).
- J. SOUSA and A. ZAGHINI, *Monetary policy shocks in the Euro Area and global liquidity spillovers,* International Journal of Finance and Economics, v.13, 3, pp. 205-218, **TD No. 629 (June 2007).**
- M. DEL GATTO, GIANMARCO I. P. OTTAVIANO and M. PAGNINI, Openness to trade and industry cost dispersion: Evidence from a panel of Italian firms, Journal of Regional Science, v. 48, 1, pp. 97-129, TD No. 635 (June 2007).
- P. DEL GIOVANE, S. FABIANI and R. SABBATINI, What's behind "inflation perceptions"? A survey-based analysis of Italian consumers, in P. Del Giovane e R. Sabbatini (eds.), The Euro Inflation and Consumers' Perceptions. Lessons from Italy, Berlin-Heidelberg, Springer, TD No. 655 (January 2008).
- R. BRONZINI, G. DE BLASIO, G. PELLEGRINI and A. SCOGNAMIGLIO, La valutazione del credito d'imposta per gli investimenti, Rivista di politica economica, v. 98, 4, pp. 79-112, TD No. 661 (April 2008).

- B. BORTOLOTTI, and P. PINOTTI, *Delayed privatization*, Public Choice, v. 136, 3-4, pp. 331-351, **TD No.** 663 (April 2008).
- R. BONCI and F. COLUMBA, *Monetary policy effects: New evidence from the Italian flow of funds*, Applied Economics, v. 40, 21, pp. 2803-2818, **TD No. 678 (June 2008).**
- M. CUCCULELLI, and G. MICUCCI, *Family Succession and firm performance: evidence from Italian family firms*, Journal of Corporate Finance, v. 14, 1, pp. 17-31, **TD No. 680 (June 2008).**
- A. SILVESTRINI and D. VEREDAS, *Temporal aggregation of univariate and multivariate time series models: a survey*, Journal of Economic Surveys, v. 22, 3, pp. 458-497, **TD No. 685 (August 2008).**

2009

- F. PANETTA, F. SCHIVARDI and M. SHUM, *Do mergers improve information? Evidence from the loan market*, Journal of Money, Credit, and Banking, v. 41, 4, pp. 673-709, **TD No. 521 (October 2004).**
- M. BUGAMELLI and F. PATERNÒ, *Do workers' remittances reduce the probability of current account reversals?*, World Development, v. 37, 12, pp. 1821-1838, **TD No. 573 (January 2006).**
- P. PAGANO and M. PISANI, *Risk-adjusted forecasts of oil prices*, The B.E. Journal of Macroeconomics, v. 9, 1, Article 24, **TD No. 585 (March 2006).**
- M. PERICOLI and M. SBRACIA, The CAPM and the risk appetite index: theoretical differences, empirical similarities, and implementation problems, International Finance, v. 12, 2, pp. 123-150, TD No. 586 (March 2006).
- U. ALBERTAZZI and L. GAMBACORTA, *Bank profitability and the business cycle*, Journal of Financial Stability, v. 5, 4, pp. 393-409, **TD No. 601 (September 2006).**
- S. MAGRI, *The financing of small innovative firms: the Italian case*, Economics of Innovation and New Technology, v. 18, 2, pp. 181-204, **TD No. 640 (September 2007).**
- V. DI GIACINTO and G. MICUCCI, The producer service sector in Italy: long-term growth and its local determinants, Spatial Economic Analysis, Vol. 4, No. 4, pp. 391-425, TD No. 643 (September 2007).
- F. LORENZO, L. MONTEFORTE and L. SESSA, *The general equilibrium effects of fiscal policy: estimates for the euro area*, Journal of Public Economics, v. 93, 3-4, pp. 559-585, **TD No. 652** (November 2007).
- R. GOLINELLI and S. MOMIGLIANO, *The Cyclical Reaction of Fiscal Policies in the Euro Area. A Critical Survey of Empirical Research*, Fiscal Studies, v. 30, 1, pp. 39-72, **TD No. 654 (January 2008).**
- P. DEL GIOVANE, S. FABIANI and R. SABBATINI, What's behind "Inflation Perceptions"? A survey-based analysis of Italian consumers, Giornale degli Economisti e Annali di Economia, v. 68, 1, pp. 25-52, TD No. 655 (January 2008).
- F. MACCHERONI, M. MARINACCI, A. RUSTICHINI and M. TABOGA, *Portfolio selection with monotone mean*variance preferences, Mathematical Finance, v. 19, 3, pp. 487-521, **TD No. 664 (April 2008).**
- M. AFFINITO and M. PIAZZA, What are borders made of? An analysis of barriers to European banking integration, in P. Alessandrini, M. Fratianni and A. Zazzaro (eds.): The Changing Geography of Banking and Finance, Dordrecht Heidelberg London New York, Springer, TD No. 666 (April 2008).
- A. BRANDOLINI, On applying synthetic indices of multidimensional well-being: health and income inequalities in France, Germany, Italy, and the United Kingdom, in R. Gotoh and P. Dumouchel (eds.), Against Injustice. The New Economics of Amartya Sen, Cambridge, Cambridge University Press, TD No. 668 (April 2008).
- G. FERRERO and A. NOBILI, *Futures contract rates as monetary policy forecasts*, International Journal of Central Banking, v. 5, 2, pp. 109-145, **TD No. 681 (June 2008).**
- P. CASADIO, M. LO CONTE and A. NERI, Balancing work and family in Italy: the new mothers' employment decisions around childbearing, in T. Addabbo and G. Solinas (eds.), Non-Standard Employment and Qualità of Work, Physica-Verlag. A Sprinter Company, TD No. 684 (August 2008).
- L. ARCIERO, C. BIANCOTTI, L. D'AURIZIO and C. IMPENNA, *Exploring agent-based methods for the analysis* of payment systems: A crisis model for StarLogo TNG, Journal of Artificial Societies and Social Simulation, v. 12, 1, **TD No. 686 (August 2008).**
- A. CALZA and A. ZAGHINI, Nonlinearities in the dynamics of the euro area demand for M1, Macroeconomic Dynamics, v. 13, 1, pp. 1-19, **TD No. 690 (September 2008).**
- L. FRANCESCO and A. SECCHI, *Technological change and the households' demand for currency*, Journal of Monetary Economics, v. 56, 2, pp. 222-230, **TD No. 697 (December 2008).**
- G. ASCARI and T. ROPELE, *Trend inflation, taylor principle, and indeterminacy*, Journal of Money, Credit and Banking, v. 41, 8, pp. 1557-1584, **TD No. 708** (May 2007).

- S. COLAROSSI and A. ZAGHINI, Gradualism, transparency and the improved operational framework: a look at overnight volatility transmission, International Finance, v. 12, 2, pp. 151-170, **TD No. 710** (May 2009).
- M. BUGAMELLI, F. SCHIVARDI and R. ZIZZA, *The euro and firm restructuring*, in A. Alesina e F. Giavazzi (eds): Europe and the Euro, Chicago, University of Chicago Press, **TD No. 716 (June 2009).**
- B. HALL, F. LOTTI and J. MAIRESSE, *Innovation and productivity in SMEs: empirical evidence for Italy*, Small Business Economics, v. 33, 1, pp. 13-33, **TD No. 718 (June 2009).**

2010

- A. PRATI and M. SBRACIA, Uncertainty and currency crises: evidence from survey data, Journal of Monetary Economics, v, 57, 6, pp. 668-681, **TD No. 446 (July 2002).**
- S. MAGRI, *Debt maturity choice of nonpublic Italian firms*, Journal of Money, Credit, and Banking, v.42, 2-3, pp. 443-463, **TD No. 574 (January 2006).**
- R. BRONZINI and P. PISELLI, *Determinants of long-run regional productivity with geographical spillovers: the role of R&D, human capital and public infrastructure,* Regional Science and Urban Economics, v. 39, 2, pp.187-199, **TD No. 597 (September 2006).**
- E. IOSSA and G. PALUMBO, *Over-optimism and lender liability in the consumer credit market*, Oxford Economic Papers, v. 62, 2, pp. 374-394, **TD No. 598 (September 2006).**
- S. NERI and A. NOBILI, *The transmission of US monetary policy to the euro area*, International Finance, v. 13, 1, pp. 55-78, **TD No. 606 (December 2006).**
- F. ALTISSIMO, R. CRISTADORO, M. FORNI, M. LIPPI and G. VERONESE, New Eurocoin: Tracking Economic Growth in Real Time, Review of Economics and Statistics, v. 92, 4, pp. 1024-1034, TD No. 631 (June 2007).
- A. CIARLONE, P. PISELLI and G. TREBESCHI, *Emerging Markets' Spreads and Global Financial Conditions*, Journal of International Financial Markets, Institutions & Money, v. 19, 2, pp. 222-239, **TD No.** 637 (June 2007).
- U. ALBERTAZZI and L. GAMBACORTA, *Bank profitability and taxation*, Journal of Banking and Finance, v. 34, 11, pp. 2801-2810, **TD No. 649** (November 2007).
- M. IACOVIELLO and S. NERI, *Housing market spillovers: evidence from an estimated DSGE model,* American Economic Journal: Macroeconomics, v. 2, 2, pp. 125-164, **TD No. 659 (January 2008).**
- F. BALASSONE, F. MAURA and S. ZOTTERI, Cyclical asymmetry in fiscal variables in the EU, Empirica, TD No. 671, v. 37, 4, pp. 381-402 (June 2008).
- F. D'AMURI, O. GIANMARCO I.P. and P. GIOVANNI, The labor market impact of immigration on the western german labor market in the 1990s, European Economic Review, v. 54, 4, pp. 550-570, TD No. 687 (August 2008).
- A. ACCETTURO, Agglomeration and growth: the effects of commuting costs, Papers in Regional Science, v. 89, 1, pp. 173-190, **TD No. 688 (September 2008).**
- S. NOBILI and G. PALAZZO, *Explaining and forecasting bond risk premiums*, Financial Analysts Journal, v. 66, 4, pp. 67-82, **TD No. 689 (September 2008).**
- A. B. ATKINSON and A. BRANDOLINI, *On analysing the world distribution of income*, World Bank Economic Review, v. 24, 1, pp. 1-37, **TD No. 701 (January 2009).**
- R. CAPPARIELLO and R. ZIZZA, Dropping the Books and Working Off the Books, Labour, v. 24, 2, pp. 139-162, **TD No. 702 (January 2009).**
- C. NICOLETTI and C. RONDINELLI, *The (mis)specification of discrete duration models with unobserved heterogeneity: a Monte Carlo study*, Journal of Econometrics, v. 159, 1, pp. 1-13, **TD No. 705** (March 2009).
- V. DI GIACINTO, G. MICUCCI and P. MONTANARO, Dynamic macroeconomic effects of public capital: evidence from regional Italian data, Giornale degli economisti e annali di economia, v. 69, 1, pp. 29-66, TD No. 733 (November 2009).
- F. COLUMBA, L. GAMBACORTA and P. E. MISTRULLI, *Mutual Guarantee institutions and small business finance*, Journal of Financial Stability, v. 6, 1, pp. 45-54, **TD No. 735** (November 2009).
- A. GERALI, S. NERI, L. SESSA and F. M. SIGNORETTI, *Credit and banking in a DSGE model of the Euro Area,* Journal of Money, Credit and Banking, v. 42, 6, pp. 107-141, **TD No. 740 (January 2010).**
- M. AFFINITO and E. TAGLIAFERRI, *Why do (or did?) banks securitize their loans? Evidence from Italy*, Journal of Financial Stability, v. 6, 4, pp. 189-202, **TD No. 741 (January 2010).**

- S. FEDERICO, *Outsourcing versus integration at home or abroad and firm heterogeneity*, Empirica, v. 37, 1, pp. 47-63, **TD No. 742 (February 2010).**
- V. DI GIACINTO, *On vector autoregressive modeling in space and time*, Journal of Geographical Systems, v. 12, 2, pp. 125-154, **TD No. 746 (February 2010).**
- S. MOCETTI and C. PORELLO, *How does immigration affect native internal mobility? new evidence from Italy*, Regional Science and Urban Economics, v. 40, 6, pp. 427-439, **TD No. 748 (March 2010)**.
- A. DI CESARE and G. GUAZZAROTTI, An analysis of the determinants of credit default swap spread changes before and during the subprime financial turmoil, Journal of Current Issues in Finance, Business and Economics, v. 3, 4, pp., **TD No. 749** (March 2010).
- A. BRANDOLINI, S. MAGRI and T. M SMEEDING, Asset-based measurement of poverty, Journal of Policy Analysis and Management, v. 29, 2, pp. 267-284, **TD No. 755** (March 2010).
- G. CAPPELLETTI, A Note on rationalizability and restrictions on beliefs, The B.E. Journal of Theoretical Economics, v. 10, 1, pp. 1-11, **TD No. 757** (April 2010).
- S. DI ADDARIO and D. VURI, *Entrepreneurship and market size. the case of young college graduates in Italy*, Labour Economics, v. 17, 5, pp. 848-858, **TD No. 775 (September 2010).**

#### FORTHCOMING

- L. MONTEFORTE and S. SIVIERO, *The Economic Consequences of Euro Area Modelling Shortcuts*, Applied Economics, **TD No. 458 (December 2002).**
- M. BUGAMELLI and A. ROSOLIA, *Produttività e concorrenza estera*, Rivista di politica economica, **TD No.** 578 (February 2006).
- G. DE BLASIO and G. NUZZO, *Historical traditions of civicness and local economic development*, Journal of Regional Science, **TD No. 591 (May 2006).**
- S. DI ADDARIO, Job search in thick markets, Journal of Urban Economics, TD No. 605 (December 2006).
- F. SCHIVARDI and E. VIVIANO, Entry barriers in retail trade, Economic Journal, TD No. 616 (February 2007).
- G. FERRERO, A. NOBILI and P. PASSIGLIA, Assessing excess liquidity in the Euro Area: the role of sectoral distribution of money, Applied Economics, **TD No. 627** (April 2007).
- P. E. MISTRULLI, Assessing financial contagion in the interbank market: maximum entropy versus observed interbank lending patterns, Journal of Banking & Finance, **TD No. 641 (September 2007).**
- Y. ALTUNBAS, L. GAMBACORTA and D. MARQUÉS, *Securitisation and the bank lending channel*, European Economic Review, **TD No. 653 (November 2007).**
- E. CIAPANNA, Directed matching with endogenous markov probability: clients or competitors?, The RAND Journal of Economics, **TD No. 665 (April 2008).**
- M. BUGAMELLI and F. PATERNÒ, *Output growth volatility and remittances*, Economica, **TD No. 673 (June 2008).**
- V. DI GIACINTO e M. PAGNINI, Local and global agglomeration patterns: two econometrics-based indicators, Regional Science and Urban Economics, **TD No. 674 (June 2008).**
- G. BARONE and F. CINGANO, Service regulation and growth: evidence from OECD countries, Economic Journal, **TD No. 675 (June 2008).**
- S. MOCETTI, *Educational choices and the selection process before and after compulsory school*, Education Economics, **TD No. 691 (September 2008).**
- P. SESTITO and E. VIVIANO, *Reservation wages: explaining some puzzling regional patterns*, Labour, **TD No. 696 (December 2008).**
- P. PINOTTI, M. BIANCHI and P. BUONANNO, *Do immigrants cause crime?*, Journal of the European Economic Association, **TD No. 698 (December 2008).**
- R. GIORDANO and P. TOMMASINO, What determines debt intolerance? The role of political and monetary institutions, Journal of Political Economy, **TD No. 700 (January 2009).**
- L. FORNI, A. GERALI and M. PISANI, *Macroeconomic effects of greater competition in the service sector:* the case of Italy, Macroeconomic Dynamics, **TD No. 706 (March 2009).**
- Y. ALTUNBAS, L. GAMBACORTA, and D. MARQUÉS-IBÁÑEZ, *Bank risk and monetary policy*, Journal of Financial Stability, **TD No. 712 (May 2009).**
- P. ANGELINI, A. NOBILI e C. PICILLO, *The interbank market after August 2007: What has changed, and why?*, Journal of Money, Credit and Banking, **TD No. 731 (ottobre 2009).**

- L. FORNI, A. GERALI and M. PISANI, *The macroeconomics of fiscal consolidations in euro area countries,* Journal of Economic Dynamics and Control, **TD No. 747 (March 2010).**
- A. DI CESARE and G. GUAZZAROTTI, An analysis of the determinants of credit default swap spread changes before and during the subprime financial turmoil, in C. V. Karsone (eds.), Finance and Banking Developments, Nova Publishers, New York., TD No. 749 (March 2010).
- G. GRANDE and I. VISCO, A public guarantee of a minimum return to defined contribution pension scheme members, Journal of Risk, **TD No. 762 (June 2010).**
- P. DEL GIOVANE, G. ERAMO and A. NOBILI, *Disentangling demand and supply in credit developments: a survey-based analysis for Italy*, Journal of Banking and Finance, **TD No. 764 (June 2010).**
- G. BARONE and S. MOCETTI, With a little help from abroad: the effect of low-skilled immigration on the female labour supply, Labour Economics, **TD No. 766 (July 2010).**
- S. MAGRI and R. PICO, *The rise of risk-based pricing of mortgage interest rates in Italy*, Journal of Banking and Finance, **TD No. 778 (October 2010).**